Author's response to reviews

Title: Predictors of Shoulder Pain and Disability Index (SPADI) and work status after 1 year in patients with subacromial shoulder pain.

Authors:

Kaia B. Engebretsen (kaiabe@medisin.uio.no)
Margreth Grotle (margreth.grotle@medisin.uio.no)
Erik Bautz-Holter (erik.bautz-holter@medisin.uio.no)
Ole Marius Ekeberg (o.m.ekeberg@medisin.uio.no)
Jens Ivar Brox (jens.ivar.brox@rikshospitalet.uio.no)

Version: 3 Date: 23 June 2010

Author's response to reviews:

BMC Musculoskeletal Disorders

Editor

Date: 23.06.10

Manuscript submission

Enclosed is the revised Full-Length Research Article entitled "Predictors of Shoulder Pain and Disability Index (SPADI) and work status after 1 year in patients with subacromial shoulder pain ". We kindly ask you to consider this again for publication in BMC Musculoskeletal Disorders.

We have addressed the comments to each of the two reviewers in a point-by-point response to the concerns.

This manuscript represents original unpublished material, also in abstract form, and does not contain any previously published material in the text, illustrations, or tables. This manuscript is not under consideration for publication elsewhere, and further, has not been posted to the Internet for public access.

The authors:
Kaia Engebretsen, (corresponding author)
Margreth Grotle,
Erik Bautz-Holter,
Ole M Ekeberg,
Jens Ivar Brox

Authors’ contributions
Kaia Engebretsen, Jens Ivar Brox, Margreth Grotle, and Erik Bautz-Holter contributed in the planning process of the present study including the study design. Kaia Engebretsen recruited the patients and performed the statistical analysis together with Ole Marius Ekeberg. Jens Ivar Brox and Margreth Grotle helped to draft the manuscript. All authors read and approved the manuscript.

Yours sincerely,

Kaia Engebretsen, research fellow
Department of Physical Medicine and Rehabilitation
Oslo University Hospital, Ullevaal
0407 Oslo, Norway.